Author's response to reviews

Title: The Matthew effect in environmental science publication: A bibliometric analysis of chemical substances in journal articles

Authors:

Philippe A Grandjean (pgrand@health.sdu.dk)
Mette L Eriksen (mleriksen@health.sdu.dk)
Ole Ellegaard (oe@bib.sdu.dk)
Johan A. Wallin (wallin@bib.sdu.dk)

Version: 2 Date: 28 September 2011

Author's response to reviews:

Thank you for the highly useful reviews.

We have revised the MS according to the reviewer suggestions and the technical advice. In the track-changes version, changes in red also include sections that have been moved.